Author's response to reviews

Title: IgG4-related kidney disease from the renal pelvis that mimicked urothelial carcinoma: A case report

Authors:

Hui Zhang (moshanghuizi@126.com)
Xinyu Ren (renxinyu7957@163.com)
Wen Zhang (lovemexieh@126.com)
Di Yang (punchzhanghui@126.com)
Ruie Feng (fengruie1@163.com)

Version: 7
Date: 20 May 2015

Author's response to reviews: see over
Title: IgG4-related kidney disease from the renal pelvis that mimicked urothelial carcinoma: A case report

Authors:

Hui Zhang (moshanghuizi@126.com)
Xinyu Ren (renxinyu7957@163.com)
Wen Zhang (lovemexieh@126.com)
Di Yang (punchzhanghui@126.com)
Ruie Feng (fengruie1@163.com)

Version: 5
Date: 5 April 2015

Author's response to reviews: see over
Dear editor,

We highly appreciate you for reviewing our paper. We have made revisions according to the comments in the revised manuscript. Besides, the patient has given the consent for information about herself to be published in BMC Urology, and we have signed the consent form. Thank you so much.

Yours sincerely,

Hui Zhang
Response to editor’s comments

EDITORIAL REQUEST

Comments to the Author:

1. Was consent to publish obtained from the patient? If not, please explain why.

Response:
Yes, the patient has given the consent for information about herself to be published in BMC Urology. And we have signed the consent form.

2. List all figure legends after the References

Response:
According to your kind suggestion, we have listed all figure legends after the References.
List of Amendments

1. Consent form: Sign the consent form.
2. Figure legends: List all figure legends after the References